

Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL100230

Key Points:

- Biological production decreases significantly in the high latitude North Atlantic in Community Earth System Model version 2 under the SSP5-8.5 scenario
- Phytolankton type dominance shifts from diatoms to small phytoplankton
- A positive feedback loop is diagnosed where changes in the physical system decrease biological production, reducing oceanic uptake of CO₂

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Boot, d.boot@uu.nl

Citation:

Boot, A., von der Heydt, A. S., & Dijkstra, H. A. (2023). Effect of plankton composition shifts in the North Atlantic on atmospheric pCO₂. *Geophysical Research Letters*, *50*, e2022GL100230. https://doi.org/10.1029/2022GL100230

Received 29 JUN 2022 Accepted 12 NOV 2022

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any readium rewinded the architecture used in

medium, provided the original work is properly cited.

Effect of Plankton Composition Shifts in the North Atlantic on Atmospheric pCO₂

A. Boot¹, A. S. von der Heydt^{1,2}, and H. A. Dijkstra^{1,2}

¹Department of Physics, Institute for Marine and Atmospheric research Utrecht, Utrecht, University, Utrecht, The Netherlands, ²Center for Complex Systems Studies, Utrecht University, Utrecht, The Netherlands

Abstract Marine carbon cycle processes are important for taking up atmospheric CO_2 thereby reducing climate change. Net primary and export production are important pathways of carbon from the surface to the deep ocean where it is stored for millennia. Climate change can interact with marine ecosystems via changes in the ocean stratification and ocean circulation. In this study we use results from the Community Earth System Model version 2 (CESM2) to assess the effect of a changing climate on biological production and phytoplankton composition in the high latitude North Atlantic Ocean. We find a shift in phytoplankton type dominance from diatoms to small phytoplankton which reduces net primary and export productivity. Using a conceptual carbon-cycle model forced with CESM2 results, we give a rough estimate of a positive phytoplankton composition-atmospheric CO_2 feedback of approximately 60 GtCO₂/°C warming in the North Atlantic which lowers the 1.5° and 2.0°C warming safe carbon budgets.

Plain Language Summary The marine carbon cycle is important for taking up carbon from the atmosphere and thereby lowering atmospheric CO_2 concentrations. One of the ways the marine carbon cycle transports carbon from the surface to the deep ocean is biological production (net primary production and export production). Once in the deep ocean, carbon can be stored for thousands of years. Biological production is dependent on environmental conditions such as nutrient availability and ocean temperature, which can be affected by increasing atmospheric CO_2 concentrations. This can lead to a positive feedback loop, where increasing CO_2 concentration decrease biological production which in turn decreases uptake of CO_2 by the ocean, effectively increasing atmospheric CO_2 concentrations. Here, we find in an Earth System Model that under a high emission scenario, biological production decreases significantly in the high latitude North Atlantic Ocean which is primarily the result of a shift in dominant phytoplankton type in this region. Larger diatoms, which are relatively efficient in exporting carbon, are replaced by small phytoplankton which are less efficient. By using a conceptual carbon cycle model, we identify a positive feedback loop where the decrease in biological production increases atmospheric CO_2 by approximately 60 GtCO₂ per degree warming.

1. Introduction

To avoid crossing tipping points in the Earth system, it is important to keep warming of our planet to a maximum of 1.5-2°C (Lenton et al., 2019). Policymakers need to know how much carbon we can still emit before we exceed this warming. However, estimates of this safe carbon budget are difficult and subject to large uncertainties because the Earth system has many processes and feedbacks that are not completely understood yet (Matthews et al., 2021).

The marine carbon pumps are currently responsible for taking up 25%–40% of anthropogenic carbon (DeVries et al., 2017; Sabine et al., 2004). It is estimated that the biological carbon pump exports approximately 11 GtC yr⁻¹ to the deep ocean (Sanders et al., 2014) and that without this export, atmospheric pCO₂ values would be 200–400 ppm higher (Henson et al., 2022; Ito & Follows, 2005). This export production (EP) is dependent on the net primary productivity (NPP). It also depends on food web dynamics and plankton composition, since different phyto- and zooplankton species have different remineralization depths (Li et al., 2009; Marinov et al., 2013; Morán et al., 2010).

Both EP and NPP are strongly dependent on temperature, ocean circulation, stratification and nutrient input (Doney et al., 2011) which can all affect phytoplankton composition. This is important because changes in phytoplankton composition can transfer through the food web affecting fish and mammals (Beaugrand, 2009; Richardson & Schoeman, 2004) which can affect fishery yields. Furthermore, phytoplankton composition also

affects EP which indirectly influences the air-sea gas exchange of CO_2 that could result in a positive feedback loop under climate change (Cabré et al., 2015). In such a feedback higher CO_2 levels change phytoplankton composition, reducing EP and thus the uptake of CO_2 by the ocean. Finally, a changing phytoplankton composition is one of the hypotheses suggested to explain (part of) the atmospheric p CO_2 (80–100 ppm) variations in the Pleistocene glacial-interglacial cycles (Archer & Maier-Reimer, 1994; Kohfeld et al., 2005), showing the potential of this feedback to affect the climate.

Though climate change is likely to affect plankton stocks, the extent is still uncertain (Osman et al., 2019). Expected effects of global warming are changes in NPP (Behrenfeld et al., 2006), reduced nutrient concentrations in the surface ocean due to stronger stratification (Bopp et al., 2001, 2005; Fu et al., 2016), and phytoplankton composition changes, the latter also due to changing co-limitation of light and nutrients (Marinov et al., 2013; C. M. Moore et al., 2013). Furthermore, on longer timescales, severe nutrient trapping in the Southern Ocean is possible, which reduces biological productivity over most of the ocean (J. K. Moore et al., 2018). However, uncertainties remain in most (model) studies. This is because the complex timing of blooms are difficult to simulate in highly seasonal regions such as the high latitudes (Martinez et al., 2011), and also because ecosystem models that have more extensive plankton dynamics show larger community composition shifts with climate change (Dutkiewicz et al., 2013; Fu et al., 2016). Comparing Coupled Model Intercomparison Project 6 (CMIP6) models with CMIP5 models, we see an increase in intermodel spread in both NPP and EP (Henson et al., 2022; Kwiatkowski et al., 2020; Tagliabue et al., 2021), highlighting the complexity of the system.

One of the regions that is projected to be affected by climate change is the North Atlantic, a region where biological processes are known to be important for gas exchange of CO_2 (Bennington et al., 2009). Model simulations under high emission scenarios project that the future North Atlantic will have relatively low warming rates, relatively high acidification rates and a (medium to large) decrease of subsurface oxygen (Bopp et al., 2013). In addition, the stratification and ocean circulation are projected to change, such as a decrease in the strength of the Atlantic Meridional Overturning Circulation (AMOC), which could possibly result in a collapse of phytoplankton and zooplankton stocks in the North Atlantic (Schmittner, 2005). These projected changes would lead to large decreases of NPP and EP (Bopp et al., 2013; Steinacher et al., 2010).

There are multiple regions in the ocean where plankton composition has been observed to shift under the influence of interannual variability and climate change. Especially the Northeast Atlantic and the North Sea are well sampled and studied regions where phytoplankton composition shifts have been observed (e.g., Allen et al., 2020; Hinder et al., 2012) and where changes in phytoplankton biomass are related to Sea Surface Temperature (SST) changes (Edwards et al., 2022; Richardson & Schoeman, 2004). An example of such a response is the increase in abundance of coccolithophores (Rivero-Calle et al., 2015; Rousseaux & Gregg, 2015). These kind of phytoplankton composition shifts are not only occurring in the North Atlantic Ocean, but also in for example, the Canadian Arctic Ocean (Blais et al., 2017) and the Northern and Equatorial Indian Ocean (Gregg et al., 2017). Also shifts in zooplankton species have been reported in for example, the North California Current (Francis et al., 2012) and the North Atlantic (Beaugrand et al., 2002). These studies show that both phyto- and zooplankton respond to changes in climate, and this can affect the future ocean carbon sink (Hilligsøe et al., 2011).

The main novel aspect of this study is to provide an estimate of the phytoplankton composition-atmospheric pCO_2 feedback (Cabré et al., 2015) using climate model simulations for a high emission scenario (SSP5-8.5). We aim to answer the following questions: How do NPP and EP respond to higher pCO_2 levels? Does this lead to a positive or a negative feedback on atmospheric CO_2 ? And lastly, what does this imply for the safe carbon budget? To investigate these issues we have used two models, the Community Earth System Model v2 (CESM2) and the Simple Carbon Project Model v1.0 (SCP-M). The CESM2 is a state-of-the-art Earth System Model, simulating processes related to the atmosphere, land, ocean and sea ice at a relatively high spatial resolution and is therefore useful to study complex processes in much detail. The SCP-M is a simple carbon cycle box model consisting of 1 atmospheric box, 2 terrestrial biosphere boxes and 7 oceanic boxes. Results from a box model like the SCP-M are, in comparison to the CESM2, easier to understand and suitable to test (higher-order) hypotheses.

2. Method

We analyze output from the Community Earth System Model v2 (CESM2, Danabasoglu et al. (2020)) simulations as used in the Coupled Model Intercomparison Project 6 (CMIP6, Eyring et al. (2016)). This model includes the CAM6 atmospheric model, the CLM land model, the CICE sea-ice model, the POP2 ocean circulation model, and the MARBL ocean biogeochemistry model. Both POP2 (Smith et al., 2010) and MARBL (Long et al., 2021) are run on a displaced Greenland pole grid at a nominal 1° horizontal resolution, with 60 non-equidistant vertical levels. In MARBL, several elemental cycles, three explicit phytoplankton functional groups (small phytoplankton, diatoms, and diazotrophs), one implicit phytoplankton group (calcifiers), and one zooplankton group are simulated. We analyze output of a single member CESM2 simulation (Danabasoglu, 2019) which is driven by greenhouse gas emissions according to the SSP5-8.5 scenario (Green et al., 2021; B. C. O'Neill et al., 2020). Since it is an emission driven case, atmospheric pCO_2 is affected by feedbacks from the land and ocean reservoirs of carbon. The simulation period is from 2015 to 2101 and analysis results are based on monthly and yearly mean data.

Generally, CESM2 is one of the best performing Earth System Models when ocean biogeochemistry is considered globally (Séférian et al., 2020). MARBL, however, has some biases due to deficiencies in the ocean circulation model (POP2). In particular, the too sluggish deep ocean circulation in POP2 causes nutrient trapping in the deep North Pacific Ocean. In MARBL, this results in low abundances of oxygen throughout the deep Pacific Ocean, which causes organic matter to be remineralized via nitrogen, negatively affecting the quality of the nitrogen cycle simulation. This large bias was fixed by adjusting the stoichiometric ratios of organic matter in the Pacific Ocean so less oxygen is consumed by organic matter remineralization; with this fix, only local biases remain (Long et al., 2021). One of these local biases is a too high chlorophyll in the North Atlantic. This could indicate that there too much phytoplankton biomass is simulated there compared to observations. This bias needs to be kept in mind in analyzing the results.

Important for this study is how the plankton groups are resolved and defined in MARBL. The three phytoplankton groups are defined in J. Moore et al. (2001) where the small phytoplankton group represent nano and pico phytoplankton. Larger phytoplankton groups are modeled as diatoms in MARBL and are therefore limited by silicate and consequently also influence the Si-cycle. The last group, the diazotrophs, are nitrogen fixers. Coccolithophores (calcifiers) are only implicitly resolved in MARBL as part of the small phytoplankton using a variable rain ratio. Per phytoplankton functional type (PFT), parameters for the growth rate, mortality and aggregation rates vary (Long et al., 2021). Only one zooplankton functional group is represented in MARBL. This group is modeled as an adaptive class, meaning that various types of zooplankton are simulated using a differential routing of grazing of the zooplankton depending (Long et al., 2021) on the prey (the three phytoplankton groups).

Ideally, we would use a suite of CESM simulations to study the feedback processes between the marine carbon cycle and atmospheric CO₂. The main component connecting the two is the air-sea gas exchange of CO₂ which is dependent on multiple tracers (i.e., temperature, salinity, dissolved inorganic carbon (DIC) and total alkalinity). These tracers all have several different sink and source terms in the ocean, making it very difficult to assess the importance of the processes simulated by CESM2. Setting up a simulation strategy to be able to study feedback processes as, for example, the effects of the EP on outgassing, is not only difficult, but also computationally expensive. Therefore, to be able to study these feedback processes in more detail, we use the Simple Carbon Project Model v1.0 (SCP-M) (C. M. O'Neill et al., 2019) as described in Supporting Information S1 including adaptations from Boot et al. (2022a). The SCP-M consists of seven ocean boxes, one atmosphere box and two terrestrial reservoirs of carbon (Figure S1 in Supporting Information S1) in which many carbon cycle processes are captured in a parameterized way. Processes in the model that affect DIC and alkalinity (Alk) are the ocean circulation, biological production, calcium carbonate production and dissolution, river fluxes, sediment fluxes and the air-sea gas exchange of CO₂ (DIC only).

3. Results

The prescribed emissions (in Pg CO₂ per year) top in the year 2085 and decrease afterward (Figure S2a in Supporting Information S1). By the year 2100, 88 GtC has been emitted into the atmosphere following the SSP5-8.5 scenario as described in Meinshausen et al. (2020). In the simulation used (Danabasoglu, 2019), the atmospheric CO₂ concentration increases from 400 ppm in 2015 to 1,069 ppm in 2100 (Figure S2b in Supporting

Information S1) under the influence of the prescribed emissions and the exchange with the land and the ocean. Part of the emitted carbon in the simulation (Figure S3a in Supporting Information S1) is taken up by the terrestrial biosphere (6%), and part by the ocean (6%). Over time, relatively less carbon is taken up by these two reservoirs, which means more remains in the atmosphere (Figure S3b in Supporting Information S1).

The changes in the global air-sea gas CO_2 exchange are shown in the upper panels of Figure 1, where a positive sign indicates CO_2 transfer into the ocean. Whereas almost the entire ocean takes up more (or gasses out less) carbon at the end of the century compared to present-day, the North Atlantic actually takes up less. This becomes even more clear when we compare globally integrated gas exchange with the gas exchange integrated over the North Atlantic. Air-sea gas exchange increases globally until the end of the century but in the North Atlantic, it starts to decrease around the year 2040 (Figure S4 in Supporting Information S1), suggesting substantial carbon cycle changes in the North Atlantic.

The response in the North Atlantic stands out for several reasons. First of all, the warming rate of SST is relatively low in the North Atlantic, and SSTs even decrease locally (Figure S5 in Supporting Information S1). Furthermore, the annual maximum mixed layer depth in the deep water formation regions around Greenland shallows (Figure S6 in Supporting Information S1). These are related to ocean circulation changes in particular a decrease in AMOC strength at 26.5°N from 17 to 10 Sv over the simulation period (Figure S7 in Supporting Information S1). We also see a stronger upper ocean stratification (Figure S8 in Supporting Information S1), where stratification is measured here by the density difference between 200 m depth and the surface (Behrenfeld et al., 2006).

There is a large decrease in both NPP and EP (at 100 m depth) in the North Atlantic region and especially in the deep water formation areas around Greenland (Figure 1). The decrease in EP cannot completely be explained by a decrease in NPP, since EP decreases more than NPP (Figure S9 in Supporting Information S1). This suggests that the PFTs in MARBL respond differently in this region to climate change. Diazotrophs do not play a role here due to temperature limitation and hence we focus on changes in diatoms and small phytoplankton. We can determine the effect of these two PFTs by using the equation (used in CESM2) NPP_i = $\mu_{ref}T_jL_iV_iP_i$. Here *i* refers to the two PFTs (small phytoplankton and diatoms), μ_{ref} is the maximum C-specific growth rate (which is the same for both PFTs) and T_f is a temperature dependent function. Furthermore, L_i is a light limitation function, V_i a nutrient limitation function, and P_i is the biomass of PFT i. Both diatoms and small phytoplankton are limited by nitrogen (N), phosphorus (P), and iron (Fe); diatoms are also limited by silicate (Si).

In the beginning of the 21st century we see that diatoms are dominant in the high latitude North Atlantic (Figure 1), whereas small phytoplankton are not very abundant. However, at the end of the century we see a change in PFT dominance as diatoms have almost completely disappeared, while the small phytoplankton NPP and biomass (Figure S13 in Supporting Information S1) have increased. The large decrease in diatom NPP decreases total NPP in this region. This shift in phytoplankton type dominance also explains why the EP decreases faster than the NPP (Figure S9 in Supporting Information S1) since diatoms are more efficient in exporting carbon than small phytoplankton. Since diatoms become less abundant, the carbon transport from the surface to the deep ocean also becomes less efficient.

To investigate why the phytoplankton composition changes in the North Atlantic, and why diatom NPP decreases, we look into what determines the production of the different PFTs. Both PFTs are generally nitrogen limited in this region, though nitrogen limitation is stronger for diatoms. Light limitation decreases by a small amount for the diatoms in the beginning of the 21st century before becoming more or less stable. Light limitation for small phytoplankton decreases throughout the entire century. The difference in light limitation explains mostly the co-limitation: diatom co-limitation of light and nutrients increases, whereas for small phytoplankton it decreases (Figure S10 in Supporting Information S1). We can also see this in the growth rate (NPP divided by biomass) of both PFTs (Figure S10e in Supporting Information S1): the growth rate of diatoms peaks around 2035 and then gradually decreases, and the growth rate of small phytoplankton keeps on increasing.

The reduced growth rate of diatoms does not completely explain the decrease in NPP and biomass, since the biomass of diatoms decreases throughout the entire period (Figure S12f in Supporting Information S1), while the growth rate of diatoms first increases (Figure S10e in Supporting Information S1). Another reason for the decrease in biomass is advective transport of diatoms out of the North Atlantic (Figure S12 in Supporting Information S1). Looking at the advective fluxes of diatom biomass over the region $(45^{\circ}-70^{\circ}N \times 270^{\circ}-0^{\circ}E)$, we can see that in the beginning of the 21st century relatively more biomass is lost due to advection over the southern

10.1029/2022GL100230

Geophysical Research Letters





Figure 1. Changes in carbon-cycle relevant quantities in the CESM2 SSP5-8.5 emission driven simulation. Top row: gas exchange in kg C m⁻² s⁻¹. Second row: NPP integrated over top 100 m in mol C m⁻² s⁻¹. Third row: EP at 100 m depth in mol C m⁻² s⁻¹. Fourth row: NPP averaged over top 100 m in mol C m⁻³ s⁻¹ for small phytoplankton. Fifth row: as fourth row but for diatoms. Note the different scaling per row; within a row, each subplot is scaled in a similar way. Left column: averages over period 2015–2030; middle column: averaged over period 2086–2101; right column: differences between the two (middle–left).

and eastern boundaries of this region which explains the decreasing NPP over this time period. Total advection decreases mostly due to reduced biomass concentrations near the southern boundary (Figures S12d and S12e in Supporting Information S1). Small phytoplankton biomass is not affected by this advection, because biomass concentrations are low in the beginning of the boundary of this region century and only increase at the southern boundary of the region when diatom biomass decreases (Figure S11 in Supporting Information S1). Eventually, due to the reduced growth rate, the biomass of diatoms does not recover. This causes the decrease in diatom NPP and explains why small phytoplankton are able to outcompete diatoms in this region utilizing the nutrients not used by diatoms anymore.

The changes in NPP and EP affect the concentrations of DIC and Alk. These two tracers affect the pCO_2 of the surface ocean, and thus the gas exchange with the atmosphere. Within CESM, it is difficult (without further simulations) to determine the effect of the reduced NPP and EP on the air-sea gas exchange and atmospheric pCO_2 as the latter quantity is determined by many other processes which cannot be separated. This effect, however, is crucial for establishing the sign of the phytoplankton composition atmospheric CO_2 feedback associated with changes in EP. To assess the feedback strength, we use the SCP-M model in combination with the CESM2 data.

Compared to the original model (C. M. O'Neill et al., 2019), we updated the SCP-M forcing files for the period 2015–2101 to represent the SSP5-8.5 scenario. We initialize the SCP-M with DIC, Alk and atmospheric CO_2 data from the CESM2 simulation of the year 2015. For the other tracers and the terrestrial biosphere initial conditions are taken from a run performed up until 2014 with historical emissions. The SCP-M captures less dynamics than the CESM2 due to reduced model complexity. We therefore first determine how large this term is to be able to separate this from the feedback strength (Figures 2e and 2f). The relative uncaptured processes by the SCP-M amounts up to approximately 8% of the total atmospheric CO_2 concentration of the CESM2 which could be caused by biases in either model. After determining the uncaptured dynamics, we estimate the feedback strength by allowing both the DIC and Alk biological fluxes to vary as a function of atmospheric pCO₂ following a fit to CESM2 output (Figures 2c and 2d). This fit represents the rate of change of DIC and Alk due to biological activity and the CESM2 output variables are averaged over the region 40° – 60° N × 270° – 30° E in the top 150 m of the water column (Figures 2a and 2b).

Using our method we can give a first estimate of the order of magnitude of the feedback. We find that the effect of reduced NPP and EP causes variability in both DIC and Alk which results in a cumulative flux of approximately 294 GtCO₂ extra in the atmosphere in the year 2100, resulting in a 37.9 ppm higher CO₂ concentration in the atmosphere (Figure 3). Over this time period global mean surface temperature rises 4.9° C (average 2096–2100 minus average 2015–2019). This process hence represents a positive feedback with a strength of 60 $\left(\frac{294}{4.9}\right)$ GtCO₂/°C warming where, due to increasing CO₂ concentrations, changes in the physical system such as an increase in stratification in the North Atlantic, result in an unfavorable environment for diatoms while small

4. Summary and Discussion

phytoplankton profit.

In this study, we investigated the interaction of atmospheric pCO_2 and biological production in the Atlantic Ocean north of 45°N in an emission driven SSP5-8.5 scenario simulation in CESM2. We found that under these high emissions, net primary production (NPP) and EP decrease in this region. Similar results for NPP and EP have been obtained in CMIP5 simulations under the RCP8.5 scenario (Bopp et al., 2013). It was shown that in the CESM2 simulation, this could be attributed to reduced productivity of diatoms which could be explained by increasing co-limitation of light and nitrogen and decreasing biomass stocks. The increasing limitation was the result of stronger stratification in the North Atlantic, which could be partly explained by increasing atmospheric temperatures due to increasing atmospheric CO_2 levels. The shift in phytoplankton composition from larger diatoms to small phytoplankton is in agreement with theory and model results suggesting that small phytoplankton outcompete diatoms under increased nutrient stress (Marinov et al., 2013). The results are also in agreement with suggestions that changes in phytoplankton composition will affect the ocean carbon sink (Hilligsøe et al., 2011).

The results indicate the existence of a positive carbon cycle feedback where plankton composition plays a central role as shown in Figure 3. Using an idealized carbon cycle model, we have made a first attempt to put an order of magnitude on the feedback strength. Obviously, this method has several caveats since the SCP-M captures less





Figure 2. (a) Change of Alk flux in the region 40° – 60° N × 270 $^{\circ}$ – 30° E in the top 150 m of the water column due to biological activity in the CESM2 versus time. (b) As in (a) but for dissolved inorganic carbon (DIC). (c) Blue markers represent CESM2 data as in (a) but versus atmospheric CO₂ concentration, and the red line represents a logarithmic fit to this data. (d) As (c) but for DIC. Note that the residuals in both (c and d) suggest systemic inadequacy in the statistical model that was fit. (e) The uncaptured dynamics for atmospheric CO₂ concentrations in ppm/yr in the SCP-M with respect to CESM2 in blue, and the feedback strength in red. (f) As in (e) but cumulative and relative to the atmospheric CO₂ concentration.

dynamics than the CESM2 and it is therefore difficult to assess the reliability of the results. In our method, the uncaptured dynamics is quite sensitive to initial conditions and parameter values. However, the actual feedback strength is only sensitive to the original strength of the biological flux.

With our method we give a first estimate of the order of magnitude of the feedback, resulting in $\sim 294 \text{ GtCO}_2$ extra in the atmosphere. To put the 294 GtCO₂ into perspective, this is slightly smaller than the increased CO₂ storage of $\sim 336 \text{ GtCO}_2$ in the North Atlantic in the Last Glacial Maximum due to a more efficient biological





Figure 3. Positive feedback loop on how biological activity in the North Atlantic Ocean is coupled to atmospheric pCO_2 . Left graph in the loop represents the increase of atmospheric pCO_2 due to the feedback loop in ppm. The right graph represents the equivalent cumulative change in air-sea gas exchange of CO_2 in GtCO₂.

carbon pump (Yu et al., 2019), meaning that the order of magnitude of our feedback strength is in a realistic range. This positive feedback (Cabré et al., 2015) seems to be relatively small, but it needs to be taken into account when estimating the safe carbon budget in future climate change. The safe carbon budget is estimated to be 308 GtCO₂ for a peak global warming of 1.5° C, and 994 GtCO₂ for a warming of 2° C (van der Ploeg, 2018). Assuming the assessed feedback strength is correct, this feedback accounts for approximately 29% and 12% of the 1.5° and 2.0° C warming safe carbon budget respectively. In summary, biological activity in the ocean is able to interact with the physical system and have an impact on variables such as global mean surface temperature via atmospheric CO₂ concentrations, with a substantial impact on our safe carbon budget. Furthermore, we want to stress that the numbers presented here should be viewed as an estimate on the order of magnitude of the feedback strength and not as an exact result because the method used to asses the feedback strength has multiple caveats.

Certainly this study has its limitations, as only a single Earth System Model with only a single member simulation for only one emission scenario is used. The biases identified in CESM2 (Long et al., 2021), especially the too large North Atlantic chlorophyll in the CESM2, can possibly affect our results. We expect that this bias increases the estimated feedback strength, because the initial phytoplankton biomass is overestimated. It might therefore be interesting to repeat such simulations and analysis with models having different phytoplankton dynamics since responses of Earth System Models depend on the complexity of this dynamics (Dutkiewicz et al., 2013; Fu et al., 2016) and since there exists a large intermodel spread in NPP and EP among CMIP6 models (Kwiatkowski et al., 2020; Tagliabue et al., 2021). Adding more zooplankton groups will also potentially affect food web dynamics in the model and can therefore influence phytoplankton composition. It is therefore interesting to see what results models with more phytoplankton and zooplankton groups, such as the trait based MARBL-SPECTRA model (Negrete-García et al., 2022), will give. Another effect of using more detailed plankton models is that more processes, which are potentially important for carbon export, are resolved. Examples of processes that are currently not resolved are jelly falls by gelatinous zooplankton which can account up to 32%-40% of the global Particulate Organic Carbon export (Luo et al., 2020), and, for example, the repackaging of fecal pellets and zooplankton that can swim (Steinberg & Landry, 2017). Though not necessarily influencing the feedback mechanism, they can affect the feedback strength because of their potentially large influence on carbon export. Furthermore, it would be useful to extend such simulations to, for example, 2300 to see whether the increased productivity of the small phytoplankton groups are able to dampen the positive feedback, or whether they will also become more limited due to increasing stratification in the North Atlantic.

Acknowledgments

This study has been supported by the

Science (OCW; Grant 024.002.001).

Netherlands Earth System Science Centre

(NESSC), which is financially supported

by the Ministry of Education, Culture and

19448007, 2023, 2, Downloaded

Data Availability Statement

The Community Earth System Model v2 (CESM2) output can be downloaded from the Earth System Grid Federation (ESGF) (https://esgf-node.llnl.gov/search/cmip6/) (Danabasoglu, 2019). Processed datasets, scripts for plotting, and scripts related to the Simple Carbon Project Model (SCP-M) used for this study can be found at https://doi.org/10.5281/zenodo.6770132 (Boot et al., 2022b). We have also included a list of datasets downloaded from the ESGF which are necessary to create the processed datasets, and the figures. The original SCP-M V1.0 can be found at https://doi.org/10.5281/zenodo.1310161 (C. M. O'Neill et al., 2018).

References

- Allen, S., Henson, S., Hickman, A., Beaulieu, C., Pc, D., & Johns, G. D. (2020). Interannual stability of phytoplankton community composition in the North-East Atlantic. *Marine Ecology Progress Series*, 655, 43–57. https://doi.org/10.3354/meps13515
- Archer, D., & Maier-Reimer, E. (1994). Effect of deep-sea sedimentary calcite preservation on atmospheric CO₂ concentration. *Nature*, 367(6460), 260–263. https://doi.org/10.1038/367260a0
- Beaugrand, G. (2009). Decadal changes in climate and ecosystems in the North Atlantic Ocean and adjacent seas. Deep Sea Research Part II: Topical Studies in Oceanography, 56(8), 656–673. https://doi.org/10.1016/j.dsr2.2008.12.022
- Beaugrand, G., Reid, P. C., Ibañez, F., Lindley, J. A., & Edwards, M. (2002). Reorganization of North Atlantic marine copepod biodiversity and climate. Science, 296(5573), 1692–1694. https://doi.org/10.1126/science.1071329
- Behrenfeld, M. J., O'Malley, R. T., Siegel, D. A., McClain, C. R., Sarmiento, J. L., Feldman, G. C., et al. (2006). Climate-driven trends in contemporary ocean productivity. *Nature*, 444(7120), 752–755. https://doi.org/10.1038/nature05317
- Bennington, V., McKinley, G. A., Dutkiewicz, S., & Ullman, D. (2009). What does chlorophyll variability tell us about export and air-sea CO₂ flux variability in the North Atlantic? *Global Biogeochemical Cycles*, 23(3), GB3002. https://doi.org/10.1029/2008GB003241
- Blais, M., Ardyna, M., Gosselin, M., Dumont, D., Bélanger, S., Tremblay, J.-É., et al. (2017). Contrasting interannual changes in phytoplankton productivity and community structure in the coastal Canadian Arctic Ocean. *Limnology & Oceanography*, 62(6), 2480–2497. https://doi. org/10.1002/lno.10581
- Boot, A., von der Heydt, A. S., & Dijkstra, H. A. (2022a). Effect of the Atlantic meridional overturning circulation on atmospheric pCO₂ variations. *Earth System Dynamics*, 13(3), 1041–1058. https://doi.org/10.5194/esd-13-1041-2022
- Boot, A., von der Heydt, A. S., & Dijkstra, H. A. (2022b). Plankton composition shift CESM2. [Dataset]. Zenodo. https://doi.org/10.5281/ zenodo.6770132
- Bopp, L., Aumont, O., Cadule, P., Alvain, S., & Gehlen, M. (2005). Response of diatoms distribution to global warming and potential implications: A global model study. *Geophysical Research Letters*, 32(19), L19606. https://doi.org/10.1029/2005GL023653
- Bopp, L., Monfray, P., Aumont, O., Dufresne, J.-L., Le Treut, H., Madec, G., et al. (2001). Potential impact of climate change on marine export production. *Global Biogeochemical Cycles*, 15(1), 81–99. https://doi.org/10.1029/1999GB001256
- Bopp, L., Resplandy, L., Orr, J. C., Doney, S. C., Dunne, J. P., Gehlen, M., et al. (2013). Multiple stressors of ocean ecosystems in the 21st century: Projections with CMIP5 models. *Biogeosciences*, 10(10), 6225–6245. https://doi.org/10.5194/bg-10-6225-2013
- Cabré, A., Marinov, I., & Leung, S. (2015). Consistent global responses of marine ecosystems to future climate change across the IPCC AR5 Earth system models. *Climate Dynamics*, *45*(5), 1253–1280. https://doi.org/10.1007/s00382-014-2374-3
- Danabasoglu, G. (2019). NCAR CESM2 model output prepared for CMIP6 C4MIP ESM-SSP585. Earth System Grid Federation. https://doi. org/10.22033/ESGF/CMIP6.7582
- Danabasoglu, G., Lamarque, J.-F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The community Earth system model version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12(2), e2019MS001916. https://doi.org/10.1029/2019MS001916
- DeVries, T., Holzer, M., & Primeau, F. (2017). Recent increase in oceanic carbon uptake driven by weaker upper-ocean overturning. *Nature*, 542(7640), 215–218. https://doi.org/10.1038/nature21068
- Doney, S. C., Ruckelshaus, M., Emmett Duffy, J., Barry, J. P., Chan, F., English, C. A., et al. (2011). Climate change impacts on marine ecosystems. Annual Review of Marine Science, 4(1), 11–37. https://doi.org/10.1146/annurev-marine-041911-111611
- Dutkiewicz, S., Scott, J. R., & Follows, M. J. (2013). Winners and losers: Ecological and biogeochemical changes in a warming ocean. Global Biogeochemical Cycles, 27(2), 463–477. https://doi.org/10.1002/gbc.20042
- Edwards, M., Beaugrand, G., Kléparski, L., Hélaouët, P., & Reid, P. C. (2022). Climate variability and multi-decadal diatom abundance in the Northeast Atlantic. *Communications Earth & Environment*, 3(1), 162. https://doi.org/10.1038/s43247-022-00492-9
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Francis, T. B., Scheuerell, M. D., Brodeur, R. D., Levin, P. S., Ruzicka, J. J., Tolimieri, N., & Peterson, W. T. (2012). Climate shifts the interaction web of a marine plankton community. *Global Change Biology*, 18(8), 2498–2508. https://doi.org/10.1111/j.1365-2486.2012.02702.x
- Fu, W., Randerson, J. T., & Moore, J. K. (2016). Climate change impacts on net primary production (NPP) and export production (EP) regulated by increasing stratification and phytoplankton community structure in the CMIP5 models. *Biogeosciences*, 13(18), 5151–5170. https://doi. org/10.5194/bg-13-5151-2016
- Green, C., Carlisle, D., O'Neill, B. C., van Ruijven, B. J., Boyer, C., & Ebi, K. (2021). Shared socioeconomic pathways (ssps) literature database, v1, 2014-2019. NASA Socioeconomic Data and Applications Center (SEDAC). https://doi.org/10.7927/hn96-9703
- Gregg, W. W., Rousseaux, C. S., & Franz, B. A. (2017). Global trends in ocean phytoplankton: A new assessment using revised ocean colour data. *Remote Sensing Letters*, 8(12), 1102–1111. https://doi.org/10.1080/2150704X.2017.1354263
- Henson, S. A., Laufkötter, C., Leung, S., Giering, S. L. C., Palevsky, H. I., & Cavan, E. L. (2022). Uncertain response of ocean biological carbon export in a changing world. *Nature Geoscience*, 15(4), 248–254. https://doi.org/10.1038/s41561-022-00927-0
- Hilligsøe, K. M., Richardson, K., Bendtsen, J., Sørensen, L.-L., Nielsen, T. G., & Lyngsgaard, M. M. (2011). Linking phytoplankton community size composition with temperature, plankton food web structure and sea–air CO₂ flux. *Deep Sea Research Part I: Oceanographic Research Papers*, 58(8), 826–838. https://doi.org/10.1016/j.dsr.2011.06.004
- Hinder, S. L., Hays, G. C., Edwards, M., Roberts, E. C., Walne, A. W., & Gravenor, M. B. (2012). Changes in marine dinoflagellate and diatom abundance under climate change. *Nature Climate Change*, 2(4), 271–275. https://doi.org/10.1038/nclimate1388

al si (10.1 (1

- Ito, T., & Follows, M. J. (2005). Preformed phosphate, soft tissue pump and atmospheric CO₂. Journal of Marine Research, 63(4), 813–839. https://doi.org/10.1357/0022240054663231
- Kohfeld, K. E., Le Quéré, C., Harrison, P. S., & Anderson, F. R. (2005). Role of marine biology in glacial-interglacial CO₂ cycles. Science, 308(5718), 74–78. https://doi.org/10.1126/science.1105375
- Kwiatkowski, L., Torres, O., Bopp, L., Aumont, O., Chamberlain, M., Christian, J. R., et al. (2020). Twenty-first century ocean warming, acidification, deoxygenation, and upper-ocean nutrient and primary production decline from CMIP6 model projections. *Biogeosciences*, 17(13), 3439–3470. https://doi.org/10.5194/bg-17-3439-2020
- Lenton, T. M., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W., & Schellnhuber, J. (2019). Climate tipping points—Too risky to bet against. *Nature*, 575(7784), 592–595. https://doi.org/10.1038/d41586-019-03595-0
- Li, W. K. W., McLaughlin, F. A., Lovejoy, C., & Carmack, E. C. (2009). Smallest Algae thrive as the Arctic Ocean Freshens. Science, 326(5952), 539. https://doi.org/10.1126/science.1179798
- Long, M. C., Moore, J. K., Lindsay, K., Levy, M., Doney, S. C., Luo, J. Y., et al. (2021). Simulations with the marine biogeochemistry library (MARBL). Journal of Advances in Modeling Earth Systems, 13(12), e2021MS002647. https://doi.org/10.1029/2021MS002647
- Luo, J. Y., Condon, R. H., Stock, C. A., Duarte, C. M., Lucas, C. H., Pitt, K. A., & Cowen, R. K. (2020). Gelatinous zooplankton-mediated carbon flows in the global oceans: A data-driven modeling study. *Global Biogeochemical Cycles*, 34(9), e2020GB006704. https://doi. org/10.1029/2020GB006704
- Marinov, I., Doney, S. C., Lima, I. D., Lindsay, K., Moore, J. K., & Mahowald, N. (2013). North-South asymmetry in the modeled phytoplankton community response to climate change over the 21st century. *Global Biogeochemical Cycles*, 27(4), 1274–1290. https://doi. org/10.1002/2013GB004599
- Martinez, E., Antoine, D., D'Ortenzio, F., & de Boyer Montégut, C. (2011). Phytoplankton spring and fall blooms in the North Atlantic in the 1980s and 2000s. Journal of Geophysical Research, 116(C11),C11029. https://doi.org/10.1029/2010JC006836
- Matthews, D. H., Tokarska, K. B., Rogelj, J., Smith, C. J., MacDougall, A. H., Haustein, K., et al. (2021). An integrated approach to quantifying uncertainties in the remaining carbon budget. *Communications Earth & Environment*, 2(1), 7. https://doi.org/10.1038/s43247-020-00064-9
- Meinshausen, M., Nicholls, Z. R. J., Lewis, J., Gidden, M. J., Vogel, E., Freund, M., et al. (2020). The shared socio-economic pathway (SSP) greenhouse gas concentrations and their extensions to 2500. Geoscientific Model Development, 13(8), 3571–3605. https://doi.org/10.5194/gmd-13-3571-2020
- Moore, C. M., Mills, M. M., Arrigo, K. R., Berman-Frank, I., Bopp, L., Boyd, P. W., et al. (2013). Processes and patterns of oceanic nutrient limitation. *Nature Geoscience*, 6(9), 701–710. https://doi.org/10.1038/ngeo1765
- Moore, J., Doney, S. C., Kleypas, J. A., Glover, D. M., & Fung, I. Y. (2001). An intermediate complexity marine ecosystem model for the global domain. Deep Sea Research Part II: Topical Studies in Oceanography, 49(1), 403–462. https://doi.org/10.1016/S0967-0645(01)00108-4
- Moore, J. K., Fu, W., Primeau, F., Britten, G. L., Lindsay, K., Long, M., et al. (2018). Sustained climate warming drives declining marine biological productivity. *Science*, 359(6380), 1139–1143. https://doi.org/10.1126/science.aao6379
- Morán, X. A. G., López-Urrutia, Á., Calvo-Díaz, A., & Li, W. K. W. (2010). Increasing importance of small phytoplankton in a warmer ocean. Global Change Biology, 16(3), 1137–1144. https://doi.org/10.1111/j.1365-2486.2009.01960.x
- Negrete-García, G., Luo, J. Y., Long, M. C., Lindsay, K., Levy, M., & Barton, A. D. (2022). Plankton energy flows using a global size-structured and trait-based model. *Progress in Oceanography*, 209, 102898. https://doi.org/10.1016/j.pocean.2022.102898
- O'Neill, B. C., Carter, T. R., Ebi, K., Harrison, P. A., Kemp-Benedict, E., Kok, K., et al. (2020). Achievements and needs for the climate change scenario framework. *Nature Climate Change*, 10(12), 1074–1084. https://doi.org/10.1038/s41558-020-00952-0
- O'Neill, C. M., Hogg, A. M., Ellwood, M. J., Eggins, S. M., & Opdyke, B. N. (2019). The [simple carbon project] model v1.0. Geoscientific Model Development, 12(4), 1541–1572. https://doi.org/10.5194/gmd-12-1541-2019
- O'Neill, C. M., Hogg, A. M., Ellwood, M. J., Opdyke, B. N., & Eggins, S. M. (2018). [simple carbon project] model (v1.0). [code]. Zenodo. https://doi.org/10.5281/zenodo.1310161
- Osman, M. B., Das, S. B., Trusel, L. D., Evans, M. J., Fischer, H., Grieman, M. M., et al. (2019). Industrial-era decline in subarctic Atlantic productivity. *Nature*, 569(7757), 551–555. https://doi.org/10.1038/s41586-019-1181-8
- Richardson, A. J., & Schoeman, D. S. (2004). Climate impact on plankton ecosystems in the northeast Atlantic. Science, 305(5690), 1609–1612. https://doi.org/10.1126/science.1100958
- Rivero-Calle, S., Gnanadesikan, A., Del Castillo, E. C., Balch, M. W., & Guikema, D. S. (2015). Multidecadal increase in North Atlantic coccolithophores and the potential role of rising CO₂. Science, 350(6267), 1533–1537. https://doi.org/10.1126/science.aaa8026
- Rousseaux, C. S., & Gregg, W. W. (2015). Recent decadal trends in global phytoplankton composition. *Global Biogeochemical Cycles*, 29(10), 1674–1688. https://doi.org/10.1002/2015GB005139
- Sabine, C. L., Feely, R. A., Gruber, N., Key, R. M., Lee, K., Bullister, J. L., et al. (2004). The oceanic sink for anthropogenic CO₂. Science, 305(5682), 367–371. https://doi.org/10.1126/science.1097403
- Sanders, R., Henson, S. A., Koski, M., De La Rocha, C. L., Painter, S. C., Poulton, A. J., et al. (2014). The biological carbon pump in the North Atlantic. *Progress in Oceanography*, 129, 200–218. https://doi.org/10.1016/j.pocean.2014.05.005
- Schmittner, A. (2005). Decline of the marine ecosystem caused by a reduction in the Atlantic overturning circulation. *Nature*, 434(7033), 628–633. https://doi.org/10.1038/nature03476
- Séférian, R., Berthet, S., Yool, A., Palmiéri, J., Bopp, L., Tagliabue, A., et al. (2020). Tracking improvement in simulated marine biogeochemistry between CMIP5 and CMIP6. Current Climate Change Reports, 6(3), 95–119. https://doi.org/10.1007/s40641-020-00160-0
- Smith, R., Jones, P. W., Briegleb, P. A., Bryan, O., Danabasoglu, G., Dennis, M. L., et al. (2010). The parallel ocean program (POP) reference manual: Ocean component of the community climate system model (CCSM).
- Steinacher, M., Joos, F., Frölicher, T. L., Bopp, L., Cadule, P., Cocco, V., et al. (2010). Projected 21st century decrease in marine productivity: A multi-model analysis. *Biogeosciences*, 7(3), 979–1005. https://doi.org/10.5194/bg-7-979-2010

Steinberg, D. K., & Landry, M. R. (2017). Zooplankton and the ocean carbon cycle. Annual Review of Marine Science, 9(1), 413–444. https:// doi.org/10.1146/annurev-marine-010814-015924

Tagliabue, A., Kwiatkowski, L., Bopp, L., Butenschön, M., Cheung, W., Lengaigne, M., & Vialard, J. (2021). Persistent uncertainties in ocean net primary production climate change projections at regional scales raise challenges for assessing impacts on ecosystem services. *Frontiers in Climate*, 3. https://doi.org/10.3389/fclim.2021.738224

van der Ploeg, F. (2018). The safe carbon budget. Climatic Change, 147(1), 47–59. https://doi.org/10.1007/s10584-017-2132-8

Yu, J., Menviel, L., Jin, Z. D., Thornalley, D. J. R., Foster, G. L., Rohling, E. J., et al. (2019). More efficient north Atlantic carbon pump during the last glacial maximum. *Nature Communications*, 10(1), 2170. https://doi.org/10.1038/s41467-019-10028-z